Author's response to reviews

Title: Assessing the activity of nonsense-mediated mRNA decay in lung cancer

Authors:

Meng Wang (mengwang55@gmail.com)
Peiwei Zhang (peiweizhang@sibs.ac.cn)
Yufei Zhu (yfzhu@sibs.ac.cn)
Xiangyin Kong (xykong@sibs.ac.cn)
Zhenguo Zhang (zhangz.sci@gmail.com)
Landian Hu (ldhu@sibs.ac.cn)

Version: 2 Date: 25 Aug 2017

Author's response to reviews:

Dear Editor,

Thank you for handling our manuscript (MGNM-D-16-00091R1) and the valuable comments from reviewers. We are pleased that our previous revision has satisfied the reviewers. In this revision, we address the remaining editing issues as follows:

1. Availability of data and materials:
We have included GEO access numbers in the section.

2. Point-by-point response:
This document has been removed; it was only served as a formatted response to reviewers.

3. Table S1:
We have changed the column header 'Age' in table S1 to 'Age at diagnosis'.

4. Clean version:
We have accepted tracked changes in all documents including the Supplementary material and switched off this function.

5. Declarations:

We reformatted or added sub-sections such as Competing interests and Consent for publication.

6. Further check:

We examined all the documents to make sure no more typos.

We are very grateful for the important issues raised by the editor, and hope that you will agree that we have addressed them and that the manuscript is now acceptable for publication. Many thanks again for help.

Sincerely,

Zhenguo Zhang, Ph.D.
University of Rochester
Department of Biology
River Campus Box 270211
Rochester, New York 14627-0211
Phone: (585) 276-2183;